Author's response to reviews

Title: Primary repair of colon injuries: clinical study of non selective approach

Authors:

Ranko G Lazovic (scp@beotel.yu)
Goran I Barisic (barisic_goran@yahoo.com)
Zoran V Krivokapic (scp@beotel.rs)

Version: 5 Date: 27 October 2010

Author's response to reviews:

In the Abstract section ISRCTN number was included.
In Methods section second sentence was changed to: Informed consent was obtained from patients or their relatives due to the severity of injuries and urgent surgical treatment.